Author’s response to reviews

Title: Real-world evidence: How pragmatic are randomized controlled trials labeled as pragmatic?

Authors:

Rafael Dal-Ré (rfdalre@gmail.com)
Perrine Janiaud (pjaniaud@stanford.edu)
John Ioannidis (jioannid@stanford.edu)

Version: 1  Date: 08 Dec 2017

Author’s response to reviews:

December 8, 2017

Anita Marinelli, PhD
BMC Medicine

Dear Dr Marinelli

Thank you for your advice and helpful suggestions on how to better structure the abstract and the manuscript as a Debate. We have made the change that you recommended and other similar changes in the revised manuscript. We hope that this would be appropriately formatted now for further evaluation.

Sincerely,

John P.A. Ioannidis, MD, DSc